Title: Allocation of Home Care Services by Municipalities in Norway: A Document Analysis

Authors:
Solrun G Holm (solrun.holm@nord.no)
Terje A. Mathisen (terje.a.mathisen@nord.no)
Torill M Sæterstrand (torill.m.saterstrand@nord.no)
Berit S. Brinchmann (berit.s.brinchmann@nord.no)

Version: 2 Date: 13 Sep 2017

Author’s response to reviews:

Solrun G. Holm

Nord University
Post box 1490
8049 Bodø
solrun.holm@nord.no

Hilary Logan
Editor
BMC Health Services Research

13th September 2017
Dear Editor,

Re: Manuscript reference BHSR-D-16-01488. Allocation of Home Care Services by Municipalities in Norway: A Document Analysis

The manuscript has been revised, and the necessary corrections are made.

We confirm that the maps depicted in Figure 1 is our own and is not subject to copyright restrictions.

Following this cover letter are the editor’s comments with our responses in italics

Sincerely,

Solrun G. Holm

Our feedback to:

Comments to the paper: "Allocation of Home Care Services by Municipalities in Norway: A Document Analysis"

Editor Comments:

1) In the Declarations section, the Ethics approval and consent to participate section is repeated. Please remove the duplicate section.

The duplicate section is removed.
2) Please include all authors’ initials in the Authors’ contributions section.

The authors’ initials are included in the Authors' contributions section.

3) Please remove the response letter before resubmitting your manuscript.

Done

4) Please state in the cover letter whether the maps depicted in Figure 1 is your own or taken from another source.

Done.